Author’s response to reviews

Title: Lights4Violence! A Quasi-Experimental Educational Intervention in six European countries to promote positive relationships among adolescents

Authors:
Carmen Vives-Cases (carmen.vives@ua.es)
MCarmen Davo-Blanes (mdavo@ua.es)
Rosario Ferrer-Cascales (rosario.ferrer@ua.es)
Belen Sanz-Barbero (bsanz@isciii.es)
Natalia Albaladejo-Blázquez (natalia.albaladejo@ua.es)
Miriam Sánchez-San Segundo (miriam.sanchez@ua.es)
Manuel Lillo-Crespo (manuel.lillo@ua.es)
Nicola Bowes (nbowes@cardiffmet.ac.uk)
Sofia Neves (asneves@ismai.pt)
Veronica Mocanu (veronica.mocanu@umfiasi.ro)
Michaela Carausu (mihaelacarausu@yahoo.com)
Jaceb Pyzalski (pyzalski@poczta.onet.pl)
Joao Forjaz (jforjaz@isciii.es)
Iwona Chmura-Rutkowska (ichmurka@amu.edu.pl)
Cristina P. Vieira (Cristina.Vieira@uab.pt)
Consuelo Corradi (c.corradi@lumsa.it)

Version: 2 Date: 19 Mar 2019

Author’s response to reviews:

PUBH-D-18-03386R1
Dear Prof Vives-Cases,

Thank you for submitting your manuscript, "Lights4Violence! A Quasi-Experimental Educational Intervention in six European countries to promote positive relationships among adolescents" (PUBH-D-18-03386R1), to BMC Public Health.

Before we can proceed, please carry out the corrections, below:

1. Before we can proceed with your submission, can you please forward FUNDING DOCUMENTATION for our records. These documents should be sent as email attachments to the following email address, bmcpublichealth@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts.

   OK. WE HAVE ALREADY SENT IT.

2. We noticed that some of your Ethical Approvals are actually Waivers, please ensure that the information in the text (the ‘Ethics approval and consent to participate’ section) reflects this.

   OK. IT IS ALREADY DONE.

3. Please add a statement to ‘Ethics approval and consent to participate’ section where you clearly state that Ethics Approvals/Waivers covered the individual schools where you performed your intervention.

   OK. IT IS ALREADY DONE.

4. We noticed that the consent forms and some of the Ethics Approval mention protecting identity of participants, however your ‘Consent for publication’ would suggest otherwise.
Could you please clarify this discrepancy? Please note that ‘Consent for publication’ refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Consent to participate is required where individual identity is protected. Please see our Editorial Policies (https://www.biomedcentral.com/getpublished/editorial-policies#ethics+and+consent) for further information.

Thank you very much for your suggestion. As you can see on page 20, we added the following: “All partners must ask parents and children for their consent to make public the resulting video capsules, short films and photos without children’s name attached to protect their identity”.

5. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section. Currently you are missing the date of Registration.

OK. WE ADDED THE FOLLOWING INFORMATION IN THE CURRENT VERSION OF THE MANUSCRIPT (see page 2):


6. Please provide page numbers in your SPIRIT Checklist rather than just ‘Done’.

OK, THIS INFORMATION IS ALREADY INCLUDED.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

*******************************************************************************

Please submit your revised manuscript by accessing the following site:

https://www.editorialmanager.com/pubh/

If you have forgotten your password, please use the 'Send Login Details' link on the login page at https://www.editorialmanager.com/pubh/. For security reasons, your password will be reset.

Please note, if your manuscript is accepted you will not be able to make any changes to the authors, or order of authors, of your manuscript once the editor has accepted your manuscript for
publication. If you wish to make any changes to authorship before you resubmit your revisions, please reply to this email and ask for a 'Request for change in authorship' form which should be completed by all authors (including those to be removed) and returned to this email address. Please ensure that any changes in authorship fulfil the criteria for authorship as outlined in BioMed Central's editorial policies (http://www.biomedcentral.com/about/editorialpolicies#authorship).

Once you have completed and returned the form, your request will be considered and you will be advised whether the requested changes will be allowed.

By resubmitting your manuscript you confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Please be aware that we may investigate, or ask your institute to investigate, any unauthorised attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

We look forward to receiving your revised manuscript before 27 Mar 2019.

Best wishes,

Eva Szunyogova, PhD

BMC Public Health

https://bmcpublichealth.biomedcentral.com/

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

------------------------Editorial Policies------------------------
Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Public Health must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcpublichealth.biomedcentral.com/)

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

THANK YOU. WE ADDED INFORMATION IN THE SECTIONS OF AVAILABILITY OF DATA AND MATERIALS AND AUTHORS’ INFORMATION IN THE NEW VERSION OF THE MANUSCRIPT.